Author's response to reviews

Title: The association between functional movement and overweight and obesity in British primary school children

Authors:

Michael J Duncan (michael.duncan@coventry.ac.uk)
Michelle Stanley (aa8396@coventry.ac.uk)
Sheila Leddington Wright (ssx398@coventry.ac.uk)

Version: 4 Date: 13 April 2013

Author's response to reviews:

Dear Editor,

Thank you for the opportunity to revise the manuscript since its change from SMARTT to BMC Sports Science, Medicine and Rehabilitation. Following on from your email we have revised the manuscript accordingly and removed or reworked the text that overlapped with our earlier paper on the same topic.

As requested we have explicitly identified how this manuscript builds on the previous paper (See Discussion Page 12 and 13 in particular). We have highlighted the changes in red font and hope the manuscript is now satisfactory. We will of course modify the manuscript further if required.

I look forward to hearing from you in due course

Sincerely

Michael J. Duncan